Author's response to reviews

Title: Factors Determining Outcome of Children Hospitalized with Severe Pneumonia

Authors:

Karalanglin K Tiewsoh (karalanglin@rediffmail.com)
Rakesh R Lodha (rakesh_lodha@hotmail.com)
Ravindra RM Pandey (rmpandey@yahoo.com)
Shobha S Broor (shobha.broor@gmail.com)
Kalaivani MK Kalaivani (mkalaivani@hotmail.com)
Sushil SK Kabra (skkabra@hotmail.com)

Version: 3 Date: 8 September 2008

Author's response to reviews:

To

The Editor

BMC Pediatrics

We are submitting revised version of our manuscript. As suggested, following changes have been made in the manuscript

1. Title page modified. The full names, institutional addresses, and e-mail addresses for all authors included and corresponding author indicated.

2. 'Competing interests' section between the Conclusions and Authors’ contributions included.

3. Authors’ contributions section before the Acknowledgements and Reference list included

4. We tried to improve the language of the manuscript.